Author's response to reviews

Title: The Cost Effectiveness of an Early Transition from Hospital to Nursing Home for Stroke Patients: Design of a comparative study [ISRCTN58135104]

Authors:

Ron WH Heijnen (R.heijnen@mumc.nl)
Silvia MAA Evers (S.Evers@BEOZ.unimaas.nl)
Trudy GDEM van der Weijden (Trudy.vanderWeijden@hag.unimaas.nl)
Martien Limburg (martienlimburg@gmail.com)
Jos MGA Schols (Jos.Schols@hag.unimaas.nl)

Version: 2 Date: 12 May 2010

Author's response to reviews:

Maastricht, 2010-05-12

Dear Editor,

In answer to the formatting changes required by you, the following revisions have been made:

Major revisions

1. Author list: The list of authors in the manuscript has been changed in accordance with the submission system.

2. Authors contributions: Text has been added describing the contribution of each of the authors in the study.

Minor revisions

1. Title: Only the first word in the title is capitalised.

2. Title: The trail number is removed from the title.

3. Email address: The format of the email addresses of the authors has been revised.

4. Tables: The text above the table title has been placed as a legend below the table.

5. Figures: The title and fig. number are removed from the image file and are now part of the manuscript (after references).

6. References: The term et al is removed from the reference list and replaced by the authors names.
7. Highlighting/tracking: All highlighting/tracking have been removed from the manuscript.

8. Table placement text: The table placement text has been removed from the manuscript.

Kind regards,

Ron Heijnen